Author's response to reviews

Title: Atypical X-linked Agammaglobulinemia caused by a Novel BTK Mutation in a selective IgM Deficiency Patient.

Authors:

Lee-Moay Lim (limleemoay@yahoo.com.tw)
Jer-Ming Chang (jemich@cc.kmu.edu.tw)
I-Fang Wang (wangifpednephro@yahoo.com.tw)
Wei-Chiao Chang (weichiao.chang@gmail.com)
Daw-Yang Hwang (910208@ms.kmuh.org.tw)
Hung-Chun Chen (chenhc@kmu.edu.tw)

Version: 2 Date: 19 March 2013

Author's response to reviews: see over
Dear Editor,

We would like to submit a manuscript in the form of *case report* to *BMC Pediatrics*, titled “*Atypical X-linked Agammaglobulinemia caused by a Novel BTK Mutation in a selective IgM Deficiency Patient*”.

In our previous manuscript, there were 7 figures but only 3 figure legends were found. Therefore, in this revision, our manuscript has been updated with legends for all the figures.

Thank you for considering our submission.

Sincerely,

Daw-yang Hwang, M.D., Ph.D.,
Division of Nephrology, Department of Medicine, Kaohsiung Medical University Hospital, 100 Tze-You First Road, Kaohsiung City 807, Taiwan
Email: 910208@ms.kmuh.org.tw
Tel: 886-7-312-1101 Ext 7351
Fax: 886-7-322-8721